Author’s response to reviews

Title: Meta-analysis of randomized controlled trials on the efficacy and safety of intracoronary administration of tirofiban for no-reflow phenomenon

Authors:

Tao Qin (feidianzhu@126.com)
Lu Xie (xielu8282@163.com)
Meng-Hua Chen (cmhnn@sina.com)

Version: 2 Date: 14 March 2013

Author’s response to reviews: see over
Dear Editor:

This is a manuscript by Tao Qin, Lu Xie and Menghua Chen entitled "Meta-analysis of randomized controlled trials on the efficacy and safety of intracoronary administration of tirofiban for no-reflow phenomenon". It is submitted to be considered for publication as a "meta-analysis" in your journal. This paper is new. Neither the entire paper nor any part of its content has been published or has been accepted elsewhere. It is not being submitted to any other journal.

We believe the paper may be of particular interest to the readers of your journal as it evaluated a new therapy for no-reflow phenomenon.

Correspondence should be addressed to Meng-Hua Chen at the following address, phone and fax number, and email address: cmhnn@sina.com, Tel: -86-771-3277186, Fax: -86-771-5350031.

Thanks very much for your attention to our paper.

Sincerely yours,

Men-Hua Chen